Author's response to reviews

Title: Estimates of the Reproduction Number for Seasonal, Pandemic, and Zoonotic Influenza: A systematic review of the literature.

Authors:

Matthew Biggerstaff (mbiggerstaff@cdc.gov)
Simon Cauchemez (simon.cauchemez@pasteur.fr)
Carrie Reed (creed1@cdc.gov)
Manoj Gambhir (manoj.gambhir@monash.edu)
Lyn Finelli (lfinelli@cdc.gov)

Version: 2 Date: 15 April 2014

Author's response to reviews: see over
April 15, 2014

Dr. Philippa Harris  
Executive Editor, *BMC Infectious Diseases*  
236 Gray’s Inn Road  
London WC1X 8HB

Dear Dr. Harris:

I have included line and page numbers in the manuscript. I apologize for the oversight.

Thank you for your consideration of this manuscript.

Sincerely,

Matthew Biggerstaff, MPH (corresponding author)  
Centers for Disease Control and Prevention  
1600 Clifton Road NE MS A-32  
Atlanta, GA 30333  
Phone: 404-639-2198  FAX: 404-639-3866  
Email: MBiggerstaff@cdc.gov